Author's response to reviews

Title: Prevalence and determinants of chronic kidney disease in rural and urban Cameroonians: A cross-sectional study

Authors:

Francois F Kaze (francoiskaze@yahoo.fr)
Diane T Meto (diane.meto@yahoo.fr)
Marie-Patrice Halle (patricehalle@yahoo.fr)
Jeanne Ngogang (jngogang@yahoo.fr)
Andre-Pascal Kengne (apkengne@yahoo.com)

Version: 5
Date: 12 July 2015

Author's response to reviews:

MS: 3748860901650201
Prevalence and determinants of chronic kidney disease in rural and urban Cameroonians: A cross-sectional study
Response to reviewer’s comments
Dear Editor,
We are grateful to the editorial office for this positive outcome of our submission to the journal. While we feel that the manuscript already incorporate all the applicable STROBE items, we have now moved text around where possible and changed headings where relevant for the sequence to read as per the STROBE checklist (a copy is attached). We also changed the declaration of interest to competing interest.
We look forward to the outcome of your assessment.
For the co-authors,
Dr Kaze